Author's response to reviews

Title: Aspirin and P2Y12 inhibition attenuate platelet-induced ovarian cancer cell invasion

Authors:

- Niamh Cooke (niamhcooke@rcsi.ie)
- Cathy Spillane (clspilla@tcd.ie)
- Orla Sheils (OSHEILS@tcd.ie)
- John O'Leary (olearyjj@tcd.ie)
- Dermot Kenny (dkenny@rcsi.ie)

Version: 6 Date: 26 August 2015

Author's response to reviews:

To whom it may concern,

Both referees are now happy with the manuscript. All of their comments and concerns have been adequately addressed.

As requested, we have

(i) Moved the Authors' contributions to the end of the manuscript to comply with the journal's format requirements
(ii) Renamed the conflict of interest disclosure to competing interests and move to the end of the manuscript.